\documentclass[twocolumn,twoside]{article}
\makeatletter\if@twocolumn\PassOptionsToPackage{switch}{lineno}\else\fi\makeatother

\usepackage{amsfonts,amssymb,amsbsy,latexsym,amsmath,tabulary,graphicx,times,xcolor}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Following additional macros are required to function some 
% functions which are not available in the class used.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\usepackage{url,multirow,morefloats,floatflt,cancel,tfrupee}
\makeatletter


\AtBeginDocument{\@ifpackageloaded{textcomp}{}{\usepackage{textcomp}}}
\makeatother
\usepackage{colortbl}
\usepackage{xcolor}
\usepackage{pifont}
\usepackage[nointegrals]{wasysym}
\urlstyle{rm}
\makeatletter

%%%For Table column width calculation.
\def\mcWidth#1{\csname TY@F#1\endcsname+\tabcolsep}

%%Hacking center and right align for table
\def\cAlignHack{\rightskip\@flushglue\leftskip\@flushglue\parindent\z@\parfillskip\z@skip}
\def\rAlignHack{\rightskip\z@skip\leftskip\@flushglue \parindent\z@\parfillskip\z@skip}

%Etal definition in references
\@ifundefined{etal}{\def\etal{\textit{et~al}}}{}


%\if@twocolumn\usepackage{dblfloatfix}\fi
\usepackage{ifxetex}
\ifxetex\else\if@twocolumn\@ifpackageloaded{stfloats}{}{\usepackage{dblfloatfix}}\fi\fi

\AtBeginDocument{
\expandafter\ifx\csname eqalign\endcsname\relax
\def\eqalign#1{\null\vcenter{\def\\{\cr}\openup\jot\m@th
  \ialign{\strut$\displaystyle{##}$\hfil&$\displaystyle{{}##}$\hfil
      \crcr#1\crcr}}\,}
\fi
}

%For fixing hardfail when unicode letters appear inside table with endfloat
\AtBeginDocument{%
  \@ifpackageloaded{endfloat}%
   {\renewcommand\efloat@iwrite[1]{\immediate\expandafter\protected@write\csname efloat@post#1\endcsname{}}}{\newif\ifefloat@tables}%
}%

\def\BreakURLText#1{\@tfor\brk@tempa:=#1\do{\brk@tempa\hskip0pt}}
\let\lt=<
\let\gt=>
\def\processVert{\ifmmode|\else\textbar\fi}
\let\processvert\processVert

\@ifundefined{subparagraph}{
\def\subparagraph{\@startsection{paragraph}{5}{2\parindent}{0ex plus 0.1ex minus 0.1ex}%
{0ex}{\normalfont\small\itshape}}%
}{}

% These are now gobbled, so won't appear in the PDF.
\newcommand\role[1]{\unskip}
\newcommand\aucollab[1]{\unskip}
  
\@ifundefined{tsGraphicsScaleX}{\gdef\tsGraphicsScaleX{1}}{}
\@ifundefined{tsGraphicsScaleY}{\gdef\tsGraphicsScaleY{.9}}{}
% To automatically resize figures to fit inside the text area
\def\checkGraphicsWidth{\ifdim\Gin@nat@width>\linewidth
	\tsGraphicsScaleX\linewidth\else\Gin@nat@width\fi}

\def\checkGraphicsHeight{\ifdim\Gin@nat@height>.9\textheight
	\tsGraphicsScaleY\textheight\else\Gin@nat@height\fi}

\def\fixFloatSize#1{}%\@ifundefined{processdelayedfloats}{\setbox0=\hbox{\includegraphics{#1}}\ifnum\wd0<\columnwidth\relax\renewenvironment{figure*}{\begin{figure}}{\end{figure}}\fi}{}}
\let\ts@includegraphics\includegraphics

\def\inlinegraphic[#1]#2{{\edef\@tempa{#1}\edef\baseline@shift{\ifx\@tempa\@empty0\else#1\fi}\edef\tempZ{\the\numexpr(\numexpr(\baseline@shift*\f@size/100))}\protect\raisebox{\tempZ pt}{\ts@includegraphics{#2}}}}

%\renewcommand{\includegraphics}[1]{\ts@includegraphics[width=\checkGraphicsWidth]{#1}}
\AtBeginDocument{\def\includegraphics{\@ifnextchar[{\ts@includegraphics}{\ts@includegraphics[width=\checkGraphicsWidth,height=\checkGraphicsHeight,keepaspectratio]}}}

\DeclareMathAlphabet{\mathpzc}{OT1}{pzc}{m}{it}

\def\URL#1#2{\@ifundefined{href}{#2}{\href{#1}{#2}}}

%%For url break
\def\UrlOrds{\do\*\do\-\do\~\do\'\do\"\do\-}%
\g@addto@macro{\UrlBreaks}{\UrlOrds}



\edef\fntEncoding{\f@encoding}
\def\EUoneEnc{EU1}
\makeatother
\def\floatpagefraction{0.8} 
\def\dblfloatpagefraction{0.8}
\def\style#1#2{#2}
\def\xxxguillemotleft{\fontencoding{T1}\selectfont\guillemotleft}
\def\xxxguillemotright{\fontencoding{T1}\selectfont\guillemotright}

\newif\ifmultipleabstract\multipleabstractfalse%
\newenvironment{typesetAbstractGroup}{}{}%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\usepackage[authoryear]{natbib}



\newif\ifnokeywords\nokeywordsfalse

\makeatletter\input{size10-pointfive.clo}\makeatother%
\definecolor{kwdboxcolor}{RGB}{242,242,242}
\usepackage[hidelinks,colorlinks=true,allcolors=blue]{hyperref}
\linespread{1}                 
\def\floatpagefraction{0.8}

\usepackage[paperheight=11.69in,paperwidth=8.26in,top=1in,bottom=1in,left=1in,right=.75in,headsep=24pt]{geometry}

\usepackage{multirow-custom}

\makeatletter

\def\hlinewd#1{%
  \noalign{\ifnum0=`}\fi\hrule \@height #1%
  \futurelet\reserved@a\@xhline}
\def\tbltoprule{\hlinewd{1pt}\\[-14pt]}
\def\tblbottomrule{\noalign{\vspace*{6pt}}\hline\noalign{\vspace*{2pt}}}
\def\tblmidrule{\hline\noalign{\vspace*{2pt}}}



\let\@articleType\@empty
\let\@journalDoi\@empty
\let\@journalVolume\@empty
\let\@journalIssue\@empty
\let\@crossMarkLink\@empty
\let\@receivedDate\@empty
\let\@acceptedDate\@empty
\let\@revisedDate\@empty
\let\@publishedDate\@empty
\let\@copyrightYear\@empty
\let\@firstPage\@empty
\def\articleType#1{\gdef\@articleType{#1}}
\def\journalDoi#1{\gdef\@journalDoi{#1}}
\def\crossMarkLink#1{\gdef\@crossMarkLink{#1}}
\def\receivedDate#1{\gdef\@receivedDate{#1}}
\def\acceptedDate#1{\gdef\@acceptedDate{#1}}
\def\revisedDate#1{\gdef\@revisedDate{#1}}
\def\publishedDate#1{\gdef\@publishedDate{#1}}
\def\copyrightYear#1{\gdef\@copyrightYear{#1}}
\def\journalVolume#1{\gdef\@journalVolume{#1}}
\def\journalIssue#1{\gdef\@journalIssue{#1}}
\def\firstPage#1{\gdef\@firstPage{#1}}
\let\@issuelogo\@empty%
\let\@issuelogoex\@empty%
\def\issuelogo#1{\gdef\@issuelogo{#1}}
\def\issuelogoex#1{\gdef\@issuelogoex{#1}}


\def\author#1{%
  \gdef\@author{%
    \hskip-\dimexpr(\tabcolsep)\hskip5pt%
    \parbox{\dimexpr\textwidth-1pt}%
    {\fontsize{11}{13}\selectfont\raggedright #1}%
  }%
}


\usepackage{pharmascope-abs}
\usepackage{caption}
\usepackage{lastpage}
\usepackage{fancyhdr}
\usepackage[noindentafter,explicit]{titlesec}
\usepackage{fontspec}

\setmainfont[%
BoldFont=cambriab.ttf,%
ItalicFont=CAMBRIAI.TTF,%
BoldItalicFont=CAMBRIAZ.TTF]{Cambria.ttf}
\lefthyphenmin = 3

\def\title#1{%
  \gdef\@title{%
      \vspace*{-40pt}%
       \ifx\@articleType\@empty\else{\fontsize{10}{12}\scshape\selectfont\hspace{8pt}\@articleType\hfill\mbox{}\par\vspace{2pt}}\fi%
       \minipage{\linewidth}
         \hrulefill\\[-0.7pt]%%{\includegraphics[width=75pt,height=50pt]{ijrps_logo.png}}\hfill
         \mbox{~}\hspace{5pt}\parbox{.1\linewidth}{\IfFileExists{images/\@issuelogo.\@issuelogoex}{\includegraphics[width=79pt,height=61pt]{images/\@issuelogo.\@issuelogoex}}{\includegraphics[width=75pt,height=50pt]{ijrps_logo.png}}}\hfill
         \fcolorbox{kwdboxcolor}{kwdboxcolor}{\parbox{.792\linewidth}{%
          \begin{center}\fontsize{17}{17}\selectfont\scshape\vskip-7pt International Journal of Research in Pharmaceutical Sciences\hfill\end{center}%
          \vspace*{-10pt}\hspace*{4pt}{\fontsize{8}{9}\selectfont Published by JK Welfare \& Pharmascope Foundation\hfill  Journal Home Page: \href{https://ijrps.com}{\color{blue}\underline{\smash{https://ijrps.com}}}}\hspace*{4pt}\mbox{}}}%
       \par\vspace*{-1pt}\rule{\linewidth}{1.3pt}%
       \endminipage%
    \par\vspace*{9.2pt}\parbox{.98\linewidth}{\linespread{.9}\raggedright\fontsize{14}{17}\selectfont #1}%
    \vspace*{-8pt}%
  }
}


\setlength{\parindent}{0pt}
\setlength{\parskip}{0.4pc plus 1pt minus 1pt}

\def\abbrvJournalTitle{Int. J. Res. Pharm. Sci.}

\fancypagestyle{headings}{%
  \renewcommand{\headrulewidth}{0pt}%
  \renewcommand{\footrulewidth}{0.3pt}
  \fancyhf{}%
  \fancyhead[R]{%
    \fontsize{9.12}{11}\selectfont\RunningAuthor,\ \abbrvJournalTitle,\ \ifx\@journalVolume\@empty X\else\@journalVolume\fi%
    \ifx\@journalIssue\@empty\else(\@journalIssue)\fi%
    ,\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}%
  }%
  \fancyfoot[LO,RE]{\fontsize{9.12}{11}\selectfont\textcopyright\ International Journal of Research in Pharmaceutical Sciences}%
  \fancyfoot[RO,LE]{\fontsize{9.12}{11}\selectfont\thepage}
}\pagestyle{headings}

                                                         
\fancypagestyle{plain}{%
  \renewcommand{\headrulewidth}{0pt}%
  \renewcommand{\footrulewidth}{0.3pt}%
  \fancyhf{}%
  \fancyhead[R]{%
    \fontsize{9.12}{11}\selectfont\RunningAuthor,\ \abbrvJournalTitle,\ \ifx\@journalVolume\@empty X\else\@journalVolume\fi%
    \ifx\@journalIssue\@empty\else(\@journalIssue)\fi%
    ,\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}%
  }%
  \fancyfoot[LO,RE]{\fontsize{9.12}{11}\selectfont\textcopyright\ International Journal of Research in Pharmaceutical Sciences}%
  \fancyfoot[RO,LE]{\fontsize{9.12}{11}\selectfont\thepage}
  \ifx\@firstPage\@empty\else\setcounter{page}{\@firstPage}\fi
}


\def\NormalBaseline{\def\baselinestretch{1.1}}
\usepackage{textcase}
\setcounter{secnumdepth}{0}

\titleformat{\section}[block]{\bfseries\boldmath\NormalBaseline\filright\fontsize{10.5}{13}\selectfont}
{\thesection}
{6pt}
{\MakeTextUppercase{#1}}
[]
\titleformat{\subsection}[block]{\bfseries\boldmath\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubsection}
{6pt}
{#1}
[]
\titleformat{\subsubsection}[block]{\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubsubsection}
{6pt}
{#1}
[]
\titleformat{\paragraph}[block]{\NormalBaseline\filright\fontsize{10.5}{10}\selectfont}
{\theparagraph}
{6pt}
{#1}
[]
\titleformat{\subparagraph}[block]{\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubparagraph}
{6pt}
{#1}
[]

\titlespacing{\section}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subsection}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subsubsection}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\paragraph}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subparagraph}{0pt}{.5\baselineskip}{.5\baselineskip}

\captionsetup[figure]{skip=1.4pt,font=bf,labelsep=colon,justification=raggedright,singlelinecheck=false}
\captionsetup[table]{skip=1.4pt,font=bf,labelsep=colon,justification=raggedright,singlelinecheck=false}

\def\bibyear#1{#1}
%\def\bibjtitle#1{#1}   %%Publisher request
\def\bibauand{}
\setlength\bibsep{3pt}
\setlength\bibhang{8pt}

\makeatother
\date{}


\usepackage{float}

\begin{document}


\def\RunningAuthor{Ayesha  Parveen et al.}
\firstPage{2914}
\articleType{Original Article}
\receivedDate{14.04.2019}
\acceptedDate{06.07.2019}
\revisedDate{02.07.2019}
\journalVolume{10}
\journalIssue{4}
\journalDoi{ijrps.v10i4.1572}
\copyrightYear{2019}
\def\authorCount{8}
\def\affCount{4}

\def\journalTitle{International Journal of Research in Pharmaceutical Sciences}

\title{Comparative anatomy of acromion process and its association with prevalence and prevention of impingement syndrome}
\author{Ayesha~~Parveen\textsuperscript{*}\textsuperscript{1},  
            Yuvaraj~~Maria~~Francis\textsuperscript{2},  
            Nivedha~V\textsuperscript{1},  
            Senthil~Kumar~S\textsuperscript{3},  
            Gunapriya~Raghunath\textsuperscript{2},  
            Ponnusamy~Kasirajan~Sankaran\textsuperscript{4},  
            Kumaresan~M\textsuperscript{2}, 
            Balaji~K\textsuperscript{2}~\\[5pt]{\textsuperscript{1}Saveetha College of Physiotherapy\unskip, Saveetha University, Chennai-600005, Tamil Nadu, India}~\\{\textsuperscript{2}Department of Anatomy\unskip, Saveetha Medical College, Saveetha University, Chennai, Tamil Nadu, India}~\\{\textsuperscript{3}Pt Ortho, Saveetha College of Physiotherapy\unskip, Saveetha University, Chennai, Tamil Nadu, India}~\\{\textsuperscript{4}Department of Anatomy \unskip, AIIMS, Mangalagiri, Vijayawada, Andhra Pradesh, India}}

\begin{abstract}
The objective of this study is to compare the morphometric features of, acromion process of humans with that of other vertebrates, and find its association with impingement syndrome. This study also aims to prevent the incidence of impingement syndrome by using various prevention techniques. For the comparative study, human scapulae were morphometrically measured and compared with the scapulae of other vertebrates. The influence of physical activities over modification of growth of acromion into various types was studied among the swimmers' population with radiographic evidence. The acromion process of 90 scapulae collected from the Anatomy Department of Saveetha Medical College and the scapula of other vertebrates like- canine, bovine, equine and apes collected from the Anatomy Department of Madras Veterinary College were studied and measured. The mean breadth of the right side and left side curved acromion is 24.26mm and 23.84mm, respectively. And for the right side and left side flat acromion it is 25.26mm and 25.36mm respectively. For canine, it is 14.87mm and 14.34mm, respectively. For bovine it is 20.33mm and 19.94mm respectively. It is concluded that apes do not have impingement syndrome due to the cranially oriented spinous and acromion process. This study also gives evidence that physical activities influence the growth of acromion into its various types, which is one of the important factors causing impingement syndrome.
\end{abstract}\def\keywordstitle{Keywords}

\begin{keywords}Impingement syndrome,\newline Acromion process,\newline canine,\newline bovine,\newline equine
\end{keywords}
\twocolumn[ \maketitle {\printKwdAbsBox}]
\makeatletter{\raggedright\textsuperscript{*}Corresponding Author\par Name:\ Ayesha~~Parveen~\\ Phone:\ 7010925601~\\  Email:\ ayeshaparveen8991@gmail.com
        \par}\vspace*{-11pt}\hrulefill\par{\fontsize{12}{14}\selectfont ISSN: 0975-7538}\par% 
        \textsc{DOI:}\ \href{https://doi.org/10.26452/\@journalDoi}{\textcolor{blue}{\underline{\smash{https://doi.org/10.26452/\@journalDoi}}}}\par%
        \vspace*{-11pt}\hrulefill\\{\fontsize{9.12}{10.12}\selectfont Production and Hosted by}\par{IJRPS |~\href{https://ijrps.com}{https://ijrps.com}}\par%
        \vspace*{-2pt}{\fontsize{9.12}{10.12}\selectfont\textcopyright\ \@copyrightYear\ $|$ All rights reserved.}\par%
        \vspace*{-11pt}\rule{\linewidth}{1.2pt}
        \makeatother
        
    
\section{Introduction}
 The acromion is a bony process on the scapula. Together with the coracoid process, it extends laterally over the shoulder joint. It is a continuation of the scapular spine and hooks over anteriorly. It articulates with the clavicle to form the acromioclavicular joint \unskip~\citep{576437:13329335}. The scapular spine, originating from the medial border of the scapula, divides the dorsal surface into a superior and an inferior quadrant namely supraspinous fossa and infraspinous fossa which gives attachment to supraspinatus and infraspinatus muscle. The spine is prolonged laterally into a flattened or curved acromion process \unskip~\citep{576437:13329346}. According to Bigliani classification, three main types of acromion morphology have been described: type-I (flat); type-II (curved); type-III (hooked). This serves as a diagnostic tool for impingement syndrome and rotator cuff tears\unskip~\citep{576437:13329332}. In these types of the acromion, the second type is found most common among the population, whereas the first type is comparatively less common, and the third type is very rare. The occurrence of rotator cuff tears is most closely associated with type III acromion \unskip~\citep{576437:13329344}. The type - III (hooked) is also called a beaked type of acromion. 

Impingement syndrome {\textendash} also called subacromial impingement or swimmer's shoulder is a clinical syndrome which occurs when the rotator cuff tendons particularly supraspinatus tendon gets irritated and inflamed as it passes through the subacromial space. This can result in pain, weakness, and loss of normal movement in the shoulder.

When the arm is raised, the subacromial space narrows, through which the supraspinatus tendon passes. Anything that causes further narrowing has the tendency to impinge on the tendon and cause an inflammatory response, resulting in impingement syndrome.

One of the major causes of impingement syndrome is variations in the shape of the acromion. It can also occur due to other causes such as subacromial spurs, repeated over arm abduction, osteoporosis, sleeping in decubitus position and smoking.

This study also includes information regarding the acromion process and rotator cuff tears in several other vertebrates. 

In the assessment of the canine model of the rotator cuff injury and repair, canine shoulder models were used to systematically investigate the factors influencing rotator cuff injury in humans \unskip~\citep{576437:13329341}. Given the similarities between human and canine musculoskeletal system, the incidence of rotator cuff injuries is similar in both canine athletes as well as humans. The cause of rotator cuff injury in agility dogs seems to be related to repeated strain, which may initiate actual degeneration of the tendon leading to shoulder joint instability. In humans and dogs, several degenerative disorders of the supraspinatus tendon have been identified, including rotator cuff tears, calcifying tendinitis, and tendinosis as a result of overuse. The degeneration of the supraspinatus tendon and bicipital tendon is thought to be a major factor in the development of medial shoulder instability in the canine population\unskip~\citep{576437:13329351}. Medial shoulder instability in canine is similar to rotator cuff injury in humans \unskip~\cite{576437:13329351}. Rotator cuff injury is often accompanied by impingement syndrome. The rotator cuff in bovines is also similar to that of humans, and the repetitive strain of their upper extremities causes impingement\unskip~\citep{576437:13329348}.

This study includes all the respective information about the acromion process, regarding its morphometric measurements, modification through evolutionary perspectives, the influence of physical activities and genes over its growth, clinical significance and prevention of impingement syndrome. The knowledge regarding the shape and various distances of the acromion process might benefit the orthopaedists during surgical repair around the shoulder joint, and the reason behind the difference in the type of acromion process that is prone to a particular disease might benefit to get a better treatment protocol. Each individual presents with variations in shoulder anatomy, overall conditioning and fitness, and degrees of shoulder laxity, which makes the precise evaluation of pathologic lesions difficult \unskip~\citep{576437:13329355}. Patients with less slope to their acromion have a propensity towards impingement because of subacromial stenosis. The impingement is due to age-related issues also. Enthesophytes are an important factor for impingement syndrome. A study stated that acromion's shape influences the rotator cuff tears size. Scapula development is modestly affected by the genetic pathways that pattern the limb\unskip~\citep{576437:13329347,576437:13329349}. And there is only rudimentary knowledge of the genes that act upstream of the blade and head development \unskip~\citep{576437:13329338}. By contrast, much information has been gathered on genes that finely pattern these structures. For example, Pax 1/ Hoxa 5 and Hoxc 6 are involved in acromion and head development, respectively \unskip~\citep{576437:13329331,576437:13329354}. Acromion has three ossification centers ( pre-acromion, Meso-acromion, and meta-acromion).\ensuremath{^{}} These three centers start ossifying during 12 years of age and fuse with each other by 16-18 years of age. The acromion normally has a secondary center of ossification which usually fuses to the rest of the acromion by the age of 25 \unskip~\citep{576437:13329333}.

We hypothesized that the acromion's complete, unrestricted, ossification is as a hooked process (type-III), but the shoulder related physical activities done during the period of ossification of acromion, like overhead arm movements, influences the growth of the bone by serving as restriction to its growth and results in the halt of its growth as a flat (type-I) or curved(type-II) process depending upon the intensity of physical activities done.
    
\section{Materials and Methods}
The acromion process of 90 scapulae, collected from the Anatomy Department of Saveetha Medical College, Thandalam, Chennai, were studied and measured with the help of digital vernier caliper, goniometer, divider, and scale.

The parameters of measurement were


  
  \begin{enumerate}
  \item \relax Length of acromion
  \item \relax Width of acromion
  \item \relax Breadth of acromion
  \item \relax Acromiocoracoid distance
  \item \relax Acromioglenoid distance
  \item \relax Angular measurements
  \end{enumerate}
  The acromiocoracoid distance is measured between the tip of the acromion process and the tip of the coracoid process, while the acromioglenoid distance is measured between the supraglenoid tubercle and tip of the acromion.

The inclination of the acromion is recorded as the angle that is formed between the tangential line of the inferior surface of the acromion and the horizontal plane. The inclination of the articular surface of the acromion is recorded as medial, lateral, and vertical to the sagittal plane.

For the comparative study, acromion process of other vertebrates was collected from the Anatomy Department of Madras Veterinary College, Chennai was studied and measured on the following parameters:



\begin{itemize}
  \item \relax 1. Length of acromion
  \item \relax 2. Width of acromion
  \item \relax 3. Breadth of acromion
\end{itemize}
  
  
  \begin{enumerate}
  \item \relax Acromio-glenoid distance (AG-1) - from supraglenoid tubercle to the tip of the acromion.
  \item \relax Acromio-glenoid distance (AG-2) - from the center of the glenoid cavity to the tip of the acromion.
  \item \relax The thickness of spine 1- at the center
  \item \relax Thickness of spine 2- near the acromion
  \end{enumerate}
  To study the correlation of physical activities and it's an association with impingement syndrome, various swimming academies were been visited, and the swimmers of various age groups and their coaches were interviewed with various questions such as:


  
  \begin{enumerate}
  \item \relax Their present age
  \item \relax Their age when they started swimming practice
  \item \relax Their regularity in swimming
  \item \relax Any shoulder related complaints throughout their swimming carrier.
  \end{enumerate}
  For further investigations, radiographic imaging was done for various people who were exposed to various forms of physical activities involving the shoulder.

Shoulder joint radiography was taken in the A-P view.

The radiograph was taken for:


  
  \begin{enumerate}
  \item \relax Professional Swimmers
  \item \relax Nonswimmers
  \item \relax Irregular swimmers
  \item \relax People exposed to very minimal activities involving shoulder 
  \item \relax People exposed to the average amount of physical activities involving shoulder
  \end{enumerate}
  \textbf{Observation}

The scapulae and acromion of canine, bovine, equine and apes were observed. The observations were compared to that of human scapulae and acromion.

In the scapula of canine (dog) the spine is placed in the middle of the dorsum, and it increases in height from above downwards to the level of the acromion process.\unskip~\citep{576437:13329345}[Figures~\ref{f-097504b15c6d} and~\ref{f-2404fcdb3ed0}] 


\bgroup
\fixFloatSize{images/6f62db33-0532-4eb7-8a4e-4d69b2e7d6b3-upicture1.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/6f62db33-0532-4eb7-8a4e-4d69b2e7d6b3-upicture1.png}{\includegraphics{images/6f62db33-0532-4eb7-8a4e-4d69b2e7d6b3-upicture1.png}}{}
\makeatother 
\caption{\boldmath {Canine Scapula - Lateral View}}
\label{f-097504b15c6d}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/a6c7c2e2-fc5e-457d-b84b-8ed2a43a8248-upicture2.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/a6c7c2e2-fc5e-457d-b84b-8ed2a43a8248-upicture2.png}{\includegraphics{images/a6c7c2e2-fc5e-457d-b84b-8ed2a43a8248-upicture2.png}}{}
\makeatother 
\caption{\boldmath {Canine Scapula - Inferior view}}
\label{f-2404fcdb3ed0}
\end{figure}
\egroup
The scapula of bovine (cattle) consists of two surfaces, three borders, and three angles. The lateral surface is wide above and narrow below, it is traversed by the scapular spine which divides the lateral surface into a cranial supraspinous fossa and caudal infraspinous fossa.\unskip~\citep{576437:13329345} The free edge of the spine is rough and tuberous in the middle and is prolonged downward to form the acromion process. The clavicle is absent, and the scapula is connected to the axial skeleton by the muscles. [Figure~\ref{f-36b40e50e897}] 


\bgroup
\fixFloatSize{images/c598f55d-4378-42cb-a83e-4a2314a7ae1e-upicture3.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/c598f55d-4378-42cb-a83e-4a2314a7ae1e-upicture3.png}{\includegraphics{images/c598f55d-4378-42cb-a83e-4a2314a7ae1e-upicture3.png}}{}
\makeatother 
\caption{\boldmath {Bovine Scapula - Lateral View}}
\label{f-36b40e50e897}
\end{figure}
\egroup
 For equine (horse), the acromion process is absent. The subscapular fossa is deeper and partly separates the two triangular rough areas in the upper third. The glenoid notch is in the anteromedial aspect of the rim. On the posterolateral aspect of the rim, it presents a tubercle for teres minor. The tuber scapula is large and placed further away from the glenoid cavity.\unskip~\citep{576437:13329345} [Figures~\ref{f-0ab0841da876} and~\ref{f-f29c45fc4831}]


\bgroup
\fixFloatSize{images/ff6abfba-9b6b-4c6e-a4d6-b794509a8ef6-upicture4.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/ff6abfba-9b6b-4c6e-a4d6-b794509a8ef6-upicture4.png}{\includegraphics{images/ff6abfba-9b6b-4c6e-a4d6-b794509a8ef6-upicture4.png}}{}
\makeatother 
\caption{\boldmath {Equine Scapula - Lateral View}}
\label{f-0ab0841da876}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/b5c7c23d-568f-4d7e-9ea1-e913d2aff8d7-upicture5.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/b5c7c23d-568f-4d7e-9ea1-e913d2aff8d7-upicture5.png}{\includegraphics{images/b5c7c23d-568f-4d7e-9ea1-e913d2aff8d7-upicture5.png}}{}
\makeatother 
\caption{\boldmath {Equine Scapula -Absence of acromion process }}
\label{f-f29c45fc4831}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/1764dd80-b10a-4677-8fb2-8e6adaa7985b-upicture6.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/1764dd80-b10a-4677-8fb2-8e6adaa7985b-upicture6.png}{\includegraphics{images/1764dd80-b10a-4677-8fb2-8e6adaa7985b-upicture6.png}}{}
\makeatother 
\caption{\boldmath {EvolutionaryModification of Scapula } a) Chimpanzee \mbox{}\protect\newline b) Gorilla \mbox{}\protect\newline c) Orang-utan \mbox{}\protect\newline d) Homo sapiens sapiens}
\label{f-222f69768660}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/011bfcea-fd26-4333-9020-a732338921f9-upicture7.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/011bfcea-fd26-4333-9020-a732338921f9-upicture7.png}{\includegraphics{images/011bfcea-fd26-4333-9020-a732338921f9-upicture7.png}}{}
\makeatother 
\caption{\boldmath {Long Acromion and Corocoid Process}}
\label{f-90889f422fa6}
\end{figure}
\egroup
\textbf{APES }- The glenoid cavity in apes has a considerably more cranial orientation. Apes like chimpanzees, orang-utans, and gibbons have a narrow shape of acromion whereas gorillas have a wide acromion [Figures~\ref{f-222f69768660} and~\ref{f-90889f422fa6}]. The spine of the scapula is almost vertically oriented in chimpanzees and orang-utans, but in gorillas, it is slanted [Figure~\ref{f-222f69768660}].

\textbf{Morphometric analysis}

Through this morphometric study of the acromion process of the scapula, it is observed that in humans the hooked type of acromion is very rare and the curved type of acromion is found most common among the population. The acromion process is absent in equines (horses) but is present in bovines (cattle) as well as canines (dogs) with somewhat similar features as that of humans Tables~\ref{tw-4459b569a963}, \ref{tw-14595727ccdc} and~\ref{tw-fa6c9eadbc93}.


\begin{table*}[!htbp]
\caption{\boldmath {Human Acromion Measurements } }
\label{tw-4459b569a963}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.2014\linewidth-2\tabcolsep}p{\dimexpr.1986\linewidth-2\tabcolsep}p{\dimexpr.20\linewidth-2\tabcolsep}p{\dimexpr.20\linewidth-2\tabcolsep}p{\dimexpr.20\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Parameter & Right side curved & Right side flat & Left side curved & Left side flat\\
\tblmidrule 
Mean length &
  39.99 &
  42.28 &
  42.21 &
  42.28\\
Mean breadth &
  24.26 &
  25.26 &
  23.84 &
  25.36\\
Mean width &
  7.55 &
  7.62 &
  7.56 &
  7.62\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Dog Acromion Measurements} }
\label{tw-14595727ccdc}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.3352\linewidth-2\tabcolsep}p{\dimexpr.3248\linewidth-2\tabcolsep}p{\dimexpr.34\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Parameter & Right side  & Left side \\
\tblmidrule 
Mean length &
  19.98 &
  19.37\\
Mean breadth &
  14.87 &
  14.34\\
Mean width &
  3.99 &
  4.23\\
AG-1 &
  35.54 &
  36.05\\
AG-2 &
  29.31 &
  29.96\\
Thickness of spine-1 &
  3.46 &
  3.55\\
Thickness of spine-2 &
  3.37 &
  3.86\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Cattle Acromion Measurements} }
\label{tw-fa6c9eadbc93}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.3317\linewidth-2\tabcolsep}p{\dimexpr.3283\linewidth-2\tabcolsep}p{\dimexpr.34\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Parameter & Right side  & Left side \\
\tblmidrule 
Mean length &
  33.11 &
  32.22\\
Mean breadth &
  20.33 &
  19.94\\
Mean width &
  12.92 &
  12.13\\
AG-1 &
  76.90 &
  75.55\\
AG-2 &
  75.65 &
  76.75\\
Thickness of spine-1 &
  16.38 &
  16.88\\
Thickness of spine-2 &
  5.02 &
  5.33\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}
\textbf{Radiographic analysis}


\bgroup
\fixFloatSize{images/bcaa7a6f-dc84-46a4-83d9-d52f9acb6125-upicture8.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/bcaa7a6f-dc84-46a4-83d9-d52f9acb6125-upicture8.png}{\includegraphics{images/bcaa7a6f-dc84-46a4-83d9-d52f9acb6125-upicture8.png}}{}
\makeatother 
\caption{\boldmath {X-Ray of a normal swimmer's shoulder}}
\label{f-00ba43147bf2}
\end{figure}
\egroup
\textbf{Normal swimmer's shoulder}

45 years old male swimmer practicing swimming from the age of 10. He was constant with his practice till the age of 30. Type I~(flat) acromion Figure~\ref{f-00ba43147bf2}.

\textbf{Non-swimmer's shoulder with impingement syndrome}

37 years old male. He was never exposed to many physical activities involving the shoulder. Works in a BPO company. He has limited ROM in shoulder elevation movements. He developed shoulder impingement due to an accident. Type III (hooked) acromion Figure~\ref{f-ba9002e17701}.


\bgroup
\fixFloatSize{images/7783f7d6-8140-458d-9486-0f227e197a02-upicture9.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/7783f7d6-8140-458d-9486-0f227e197a02-upicture9.png}{\includegraphics{images/7783f7d6-8140-458d-9486-0f227e197a02-upicture9.png}}{}
\makeatother 
\caption{\boldmath {X-Ray of a non swimmer's shoulder with impingement syndrome }}
\label{f-ba9002e17701}
\end{figure}
\egroup
\textbf{Irregular swimmer's shoulder with impingement syndrome}

19 years old male practicing swimming irregularly from the age of 13. He is exposed to many sports, including basketball, valley ball, badminton, and swimming, etc. from the age of 12. Type II (curved) Figure~\ref{f-ba62201682d4}.

\textbf{ Normal non-swimmer' shoulder}

X-ray of a 20 years old female who is exposed to an average amount of physical activities involving the use of overhead arm movements during the age of 12-18. Type II (curved) Figure~\ref{f-78225fc3af61}.
\bgroup
\fixFloatSize{images/d4e56b86-34eb-4adc-869c-bcf81f87a86f-upicture10.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/d4e56b86-34eb-4adc-869c-bcf81f87a86f-upicture10.png}{\includegraphics{images/d4e56b86-34eb-4adc-869c-bcf81f87a86f-upicture10.png}}{}
\makeatother 
\caption{\boldmath {X-Ray of an irregular swimmer's shoulder with impingement syndrome}}
\label{f-ba62201682d4}
\end{figure}
\egroup

    
\section{Results and Discussion}
 The mean breadth of the right side and left side curved acromion is 24.26 mm and 23.84 mm respectively and for the right side and left side flat acromion it is 25.26 mm and 25.36 mm respectively. In canine, the mean breadth of the right side and left side acromion is 14.87 and 14.34 mm respectively. For bovine, the mean breadth of the right side and left side acromion is 20.33 and 19.94 mm respectively.


\bgroup
\fixFloatSize{images/4df9a9cd-9b03-43f5-af0c-816e619a2aa7-upicture11.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/4df9a9cd-9b03-43f5-af0c-816e619a2aa7-upicture11.png}{\includegraphics{images/4df9a9cd-9b03-43f5-af0c-816e619a2aa7-upicture11.png}}{}
\makeatother 
\caption{\boldmath {X-Ray of a normal non swimmer' shoulder }}
\label{f-78225fc3af61}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/798ee460-95c4-4842-9fb0-b5d3b1d34102-upicture12.png}
\begin{figure*}[!htbp]
\centering \makeatletter\IfFileExists{images/798ee460-95c4-4842-9fb0-b5d3b1d34102-upicture12.png}{\includegraphics{images/798ee460-95c4-4842-9fb0-b5d3b1d34102-upicture12.png}}{}
\makeatother 
\caption{\boldmath {shoulder abduction}}
\label{f-55593a1f586e}
\end{figure*}
\egroup

\bgroup
\fixFloatSize{images/c8e0e63b-f934-4387-a69f-bbebd69fc258-upicture13.png}
\begin{figure*}[!htbp]
\centering \makeatletter\IfFileExists{images/c8e0e63b-f934-4387-a69f-bbebd69fc258-upicture13.png}{\includegraphics{images/c8e0e63b-f934-4387-a69f-bbebd69fc258-upicture13.png}}{}
\makeatother 
\caption{\boldmath {Rows}}
\label{f-c733a6d7a629}
\end{figure*}
\egroup

\bgroup
\fixFloatSize{images/a506720a-ddb8-4dac-b710-06da3d1e20ed-upicture14.png}
\begin{figure*}[!htbp]
\centering \makeatletter\IfFileExists{images/a506720a-ddb8-4dac-b710-06da3d1e20ed-upicture14.png}{\includegraphics{images/a506720a-ddb8-4dac-b710-06da3d1e20ed-upicture14.png}}{}
\makeatother 
\caption{\boldmath {Bear Hugs}}
\label{f-2bdd2efa49b2}
\end{figure*}
\egroup

\bgroup
\fixFloatSize{images/8ef04287-96fc-4b7b-879b-d9e53fd7c5a9-upicture15.png}
\begin{figure*}[!htbp]
\centering \makeatletter\IfFileExists{images/8ef04287-96fc-4b7b-879b-d9e53fd7c5a9-upicture15.png}{\includegraphics{images/8ef04287-96fc-4b7b-879b-d9e53fd7c5a9-upicture15.png}}{}
\makeatother 
\caption{\boldmath {Biceps Curl}}
\label{f-06786cb131bb}
\end{figure*}
\egroup

\bgroup
\fixFloatSize{images/db83bc8a-9e77-4116-ab89-d0b4f0012314-upicture16.png}
\begin{figure*}[!htbp]
\centering \makeatletter\IfFileExists{images/db83bc8a-9e77-4116-ab89-d0b4f0012314-upicture16.png}{\includegraphics{images/db83bc8a-9e77-4116-ab89-d0b4f0012314-upicture16.png}}{}
\makeatother 
\caption{\boldmath {Internal rotation against resistance}}
\label{f-c27b4a4689a5}
\end{figure*}
\egroup

\bgroup
\fixFloatSize{images/fd8a0635-4cf4-42f3-b2e3-4068be779c64-upicture17.png}
\begin{figure*}[!htbp]
\centering \makeatletter\IfFileExists{images/fd8a0635-4cf4-42f3-b2e3-4068be779c64-upicture17.png}{\includegraphics{images/fd8a0635-4cf4-42f3-b2e3-4068be779c64-upicture17.png}}{}
\makeatother 
\caption{\boldmath {Internal rotation using thera band}}
\label{f-cd8306318f59}
\end{figure*}
\egroup

\bgroup
\fixFloatSize{images/dd67be48-9028-4073-a4ee-0960d4b57a47-upicture18.png}
\begin{figure*}[!htbp]
\centering \makeatletter\IfFileExists{images/dd67be48-9028-4073-a4ee-0960d4b57a47-upicture18.png}{\includegraphics{images/dd67be48-9028-4073-a4ee-0960d4b57a47-upicture18.png}}{}
\makeatother 
\caption{\boldmath {External rotation against resistance}}
\label{f-c763f2aced39}
\end{figure*}
\egroup

\bgroup
\fixFloatSize{images/b53fe7fb-17ab-48ff-b727-29a461722b45-upicture19.png}
\begin{figure*}[!htbp]
\centering \makeatletter\IfFileExists{images/b53fe7fb-17ab-48ff-b727-29a461722b45-upicture19.png}{\includegraphics{images/b53fe7fb-17ab-48ff-b727-29a461722b45-upicture19.png}}{}
\makeatother 
\caption{\boldmath {External rotation using thera band}}
\label{f-5bdedac19cef}
\end{figure*}
\egroup

\bgroup
\fixFloatSize{images/d9e2e258-5a94-4595-8733-b1b61217d2b2-upicture20.png}
\begin{figure*}[!htbp]
\centering \makeatletter\IfFileExists{images/d9e2e258-5a94-4595-8733-b1b61217d2b2-upicture20.png}{\includegraphics{images/d9e2e258-5a94-4595-8733-b1b61217d2b2-upicture20.png}}{}
\makeatother 
\caption{\boldmath {Scaption}}
\label{f-375baef65573}
\end{figure*}
\egroup

\bgroup
\fixFloatSize{images/89889ba9-9e82-4274-a5bb-a371e3d34765-upicture21.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/89889ba9-9e82-4274-a5bb-a371e3d34765-upicture21.png}{\includegraphics{images/89889ba9-9e82-4274-a5bb-a371e3d34765-upicture21.png}}{}
\makeatother 
\caption{\boldmath {Posterior deltoid stretch}}
\label{f-eb37f3cd3e15}
\end{figure}
\egroup
In canine, the mean thickness of the spine-1 for the right side and the left side is 3.46 mm and 3.55 mm respectively, and in bovine it is 16.38 mm and 16.88 mm respectively. The mean thickness of the spine-2 of canine for the right side and the left side is 3.37 and 3.86 respectively, and for bovine it is 5.02 mm for the right side and 5.33 mm for the left side.

For canine, the mean AG-1 of the right side and left side is 35.54 mm and 36.05 mm respectively and for bovine the right side and left side mean AG-1 is 76.90 mm and 75.55 mm respectively. For canine, the mean AG-2 of the right side and left side is 29.31 mm and 29.96 mm respectively, whereas in bovine it is 75.65 mm and 76.75 mm respectively.

Shoulder impingement syndrome, also called subacromial impingement, painful arc syndrome, supraspinatus syndrome, swimmer's shoulder, and thrower's shoulder is a clinical syndrome which occurs when the tendons of the rotator cuff muscles become irritated and inflamed as they pass through the subacromial space, the passage beneath the acromion. This can result in pain, weakness, and loss of movement at the shoulder\unskip~\citep{576437:13329336}.

\textbf{There are two types of subacromial impingement}

\textbf{Structural impingement}

\textbf{\space }Impingement due to a reduction of the subacromial space. The structural impingement can be caused by bone growth, inflammation of tendons, osteophytes or calcifications.

\textbf{Functional impingement}

Can be caused by glenohumeral instability and muscle imbalance. Functional impingement is a secondary consequence\unskip~\citep{576437:13329343}.

The morphometric study of the acromion is very important as the acromion process shows much variation in its morphology and also has various types that depict a great connection to impingement syndrome \unskip~\citep{576437:13329345}.

In this study and most of the other morphometric studies done by Susmita Saha et al., Paraskevas et al., Singh et al. it is observed that the type II (curved) of the acromion is most common whereas the type III (hooked) acromion is less common. In this study, the overall ratio of type II (curved): type I~(flat) acromion was 3:1. The type III (hooked) acromion was not found in this study.

A few previous studies have investigated the association between factors such as age, sex, body mass index (BMI) or smoking status with this condition. Smoking has been proven to be a preventable risk factor associated with several health conditions such as chronic lung disease, cardiovascular disease, and low back pain. This could be related to the fact that nicotine can affect sensory thresholds, impair vascularisation to tendons and disturb tendon healing capacity. Other than these factors osteoporosis, nutritional deficiency, and the decubitus sleeping position can also be one of the causes of impingement syndrome. Sleeping in decubitus position showed a 3.7 times greater risk compared with sleeping in the supine position. Repetitive overload from bodyweight during sleeping might cause the impingement of the tendon against the acromial arch, resulting in tendon degeneration and inflammation\unskip~\citep{576437:13329352}.

Osteoporosis is a disease that causes thinning of bone and tissues and can affect the shoulder bones resulting in bone pain and an increased risk for fractures. Vitamin D helps in the absorption and utilization of Calcium to keep bones and connective tissue strong. Therefore, the deficiency of vitamin D causes osteoporosis, which in turn can cause impingement syndrome. 

This study gives an idea about the morphometric as well as comparative information of human and vertebrate acromion process and its relation with impingement syndrome. We observed the scapulae of canine, bovine and equine, and found that equines do not have an acromion process and the morphological features of canine scapula were highly similar to that of humans. Due to this similarity, canines were used as models to study about rotator cuff injury in humans. In canines supraspinatus tendinopathy occurs due to repetitive activities such as quick turns, jump-turn combinations, slipping on wet surface and weave poles. This places the shoulder near its end range of abduction, which in turn leads to medial shoulder instability \unskip~\citep{576437:13329350}. Medial shoulder instability of canines is similar to rotator cuff injury in humans, which is often accompanied by impingement syndrome. In a study, a total of 327 dogs were diagnosed with supraspinatus tendinopathy from the year 2006-2013. In a study, Ronald Ailsby says, large animal veterinarians face acute and chronic repetitive strain injuries daily to their upper extremities \unskip~\citep{576437:13329349}

In a comparative study, the rotator cuff muscle architecture of animals such as dogs, rats, pigs, sheep, cows, and primates was compared with humans. It was found that the architecture of dogs and primates was more similar to that of humans \unskip~\citep{576437:13329342}.

This comparative study might help the veterinary doctors as well as orthopaedists to adapt a better treatment protocol for shoulder related problems in animals and impingement syndrome in humans. 

During the study of evolutionary perspectives of the acromion, we found that throughout man evolution, the scapula as a whole has modified to a great extent. Based on the activities done by early man and modern man, the spine of scapula had modified from almost vertical to almost horizontal in position, and so is the case with the acromion.

Since scapulae of apes were not available, we referred some of the previous studies regarding the human acromion viewed from an evolutionary perspective and observed the scapula of chimpanzee, gorilla, and orang-utan. It is found that the spine of the scapula is vertical in chimpanzee and slanted in gorilla and orang-utan. And the acromion is narrow, the glenoid cavity is facing cranially, and it is oval in all the apes\unskip~\citep{576437:13329339}.

Another previous study done by J.M. Potadu et al. on subacromial space in African great apes has proved that the acromion was more curved, and the occurrence of the impingement syndrome in apes was not possible \unskip~\citep{576437:13329340}.

For more evidence regarding absence of impingement syndrome in apes Arignar Anna Zoological Park, Vandalur, Chennai was visited, and the wildlife veterinarians were interviewed, and it was concluded that impingement syndrome was never found in apes and that their shoulder abnormalities were very rare due to their functionally and genetically modified scapula and acromion. Based on the theoretical evidence and wildlife veterinarian's experience and practical knowledge it was concluded that apes do not have shoulder impingement due to the vertical orientation of the acromion process along the spine of the scapula and its curved and narrowed feature. 

For the investigations of impingement syndrome among the swimmers' population, various swimming academies were been visited, and the swimmers of various age groups and their coaches were interviewed with various questions regarding there starting age of swimming practice, its regularity and any experience of shoulder related pain or disability. The question session gave a clear idea for supporting our hypothesis. Other than swimmers, we also went through a case of shoulder impingement in a fisherman who was practicing his profession for 20 years. He was subjected to a repetitive cycle of abduction, flexion, and extension during sailing his boat. This case also gave evidence supporting our hypothesis.

The influence of physical activities on the growth of acromion into its various types was studied with the help of radiographs of some swimmers and nonswimmers who were subjected to different sets and forms of physical activities during the ossification and fusion stage of the acromion. These radiographs were compared with each other and correlated to the amount and type of shoulder related physical activities done by them and with the radiographic evidence it was concluded that physical activities of shoulder done during the ossifying and fusing stages of the 4 ossification centers of acromion, which is between 12-25 years, does influence the modification of acromion into its various types. Excessive physical activities involving shoulder done during 12-25 years of age provides more restriction to the bone growth, and hence the acromion stops it's growth by flat or curved process and does not grow to its complete form as a hooked process.

\textbf{Prevention}

Impingement syndrome can be prevented by strengthening the rotator cuff muscles and other shoulder muscles by exercises [Figures~\ref{f-78225fc3af61}, \ref{f-55593a1f586e}, \ref{f-c733a6d7a629}, \ref{f-2bdd2efa49b2}, \ref{f-06786cb131bb}, \ref{f-c27b4a4689a5}, \ref{f-cd8306318f59}, \ref{f-c763f2aced39}, \ref{f-5bdedac19cef}, \ref{f-375baef65573} and~\ref{f-eb37f3cd3e15}] 

Exercise techniques used to strengthen these muscles are


  
  \begin{enumerate}
  \item \relax Active movements without resistance 
  \item \relax Exercises against resistance
  \item \relax Thera band exercises
  \item \relax Shoulder stabilization exercises
  \end{enumerate}
  Impingement syndrome can also be prevented by encouraging shoulder movements involving activities and sports like swimming, throwball, basketball, etc., that involves a lot of overhead arm movements, during 12-18 years of age. As this age involves the ossification of the acromion, activities done during this age will modify the ossification of bone into a type that will easily allow overhead arm movements without serving many restrictions to the movement.

This kind of practice will modify the acromion process's ossification into type I~(flat) or type II (curved), which are not prone to develop impingement syndrome unless the shoulder experiences an excessive trauma. And it also prevents the growth of the bone into type III (hooked), which is the most prone type of acromion to develop impingement syndrome.
    
\section{Conclusion}
Shoulder impingement does not occur in apes as their glenoid cavity is more cranially oriented and their spine of scapula and acromion process is almost vertical in its orientation, which provides unrestricted over-head arm movements. Whereas humans have a wide, horizontally oriented acromion above a laterally oriented glenoid cavity which provides a considerable amount of restriction to overhead abduction activities. Thus, humans have developed impingement syndrome due to these modifications during evolution. Besides the type-III (hooked) acromion being more prone to develop impingement syndrome the type II (curved) acromion though rarely but can also develop impingement syndrome in cases of trauma. Physical activities done during the ossifying and fusing stage of acromion does influence the modification of the bone into its three various types, hence exercises given during the ossifying stage of acromion can prevent its growth as type III (hooked) acromion which is most prone to develop impingement, being one of the major causes of prevalence of impingement syndrome.
\section*{Acknowledgement} This research was supported by Saveetha Medical College, Saveetha College of Physiotherapy, Madras Veterinary College and Waves Swim Academy. We are grateful to these institutions and the Academy for their support and help that made our research possible. We are also thankful to our colleagues, who provided the expertise that greatly assisted the research. There are no conflicts of interest.

\textbf{Author contributions}

Ayesha parveen - Contributed in, developing the concept of the study, formulating the hypothesis, data collection, analysis and interpretation and drafting the article. 

V.nivedha - Contributed to the acquisition of data, building the study design and drafting the manuscript.

Yuvaraj maria francis - Contributed with ideas for the progression of the study, the suggestion of topic and critical revision of the manuscript.

Senthil kumar - Contributed in the final revision of the article and its approval and guided in the submission of the article in the journal.
    

\bibliographystyle{pharmascope_apa-custom}

\bibliography{\jobname}
\end{document}
